

Global **Biogeochemical Cycles**[®]

RESEARCH ARTICLE

10.1029/2021GB007278

Key Points:

- Watershed size imposes a primary control on the Damköhler number (Da) for dissolved organic carbon (DOC) uptake in global river networks
- The Da for DOC uptake in tropical river networks is 2-6 fold that in temperate and the Arctic river networks
- Damming and climatic warming significantly enhance the Da for DOC uptake in global river networks

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

S. Liu, liushaoda@bnu.edu.cn

Citation:

Liu, S., Maavara, T., Brinkerhoff, C. B., & Raymond, P. A. (2022). Global controls on DOC reaction versus export in watersheds: A Damköhler number analysis. Global Biogeochemical Cycles, 36, e2021GB007278. https://doi. org/10.1029/2021GB007278

Received 15 DEC 2021 Accepted 1 APR 2022

Global Controls on DOC Reaction Versus Export in Watersheds: A Damköhler Number Analysis

Shaoda Liu^{1,2} , Taylor Maavara^{2,3}, Craig B. Brinkerhoff⁴, and Peter A. Raymond²

¹State Key Laboratory for Water Environment Simulation, School of Environment, Beijing Normal University, Beijing, China, ²Yale School of the Environment, Yale University, New Haven, CT, USA, ³Yale Institute for Biospheric Studies, Yale University, New Haven, CT, USA, ⁴Department of Civil and Environmental Engineering, University of Massachusetts Amherst, Amherst, MA, USA

Abstract The relative capacity for watersheds to eliminate or export reactive constituents has important implications on aquatic ecosystem ecology and biogeochemistry. Removal efficiency depends on factors that affect either the reactivity or advection of a constituent within river networks. Here, we characterized Damköhler number (Da) for dissolved organic carbon (DOC) uptake in global river networks. Da equals the advection to reaction timescale ratio and thus provides a unitless indicator for DOC reaction intensity during transport within river networks. We aim to demonstrate the spatial and temporal patterns and interplays among factors that determine DOC uptake across global river networks. We show that watershed size imposes a primary control on river network DOC uptake due to a three orders of magnitude difference in water residence time (WRT) between the smallest and largest river networks. DOC uptake capacity in tropical river networks is 2-6 times that in temperate and the Arctic river networks, coinciding with larger DOC removals in warm than in cold watersheds. River damming has a profound impact on DOC uptake due to significantly extended WRTs, particularly in temperate watersheds where most constructed dams are situated. Global warming is projected to increase river network DOC uptake by ca. 19% until year 2100 under the RCP4.5 scenario.

Plain Language Summary Dissolved organic carbon (DOC) transported by river networks is either removed during transport or exported to downstream receiving basins. The capacity for river networks to remove DOC is determined by the relative balance between its transport and reaction timescales. Damköhler number measures the transport to a reaction timescale ratio and provides a common framework for evaluating the spatial and temporal variations in DOC uptake across global river networks. We show that large river networks have higher DOC uptake because of orders of magnitude longer advection time than in small river networks. Tropical river networks have higher DOC uptake than temperate and the Arctic river networks because of higher reaction rates in warm than in cold watersheds. Damming significantly prolongs water advection and increases DOC uptake within river networks. Global warming is also projected to increase river network DOC uptake because of enhanced DOC uptake rates under a warmer climate.

1. Introduction

Terrestrial materials transported by river systems are either exported to terminal receiving basins or removed during downstream transport (Cole et al., 2007; Ensign & Doyle, 2006) via processes, including particulate sedimentation, in situ photo-/biological transformation, or surface degassing. The relative balance between within-river removal and export of these materials has important implications for biogeochemistry and ecology in both rivers (Battin et al., 2008; Raymond et al., 2013) and downstream receiving water bodies (Bauer et al., 2013; Maavara, Akbarzadeh, & Van Cappellen, 2020). The temporal dimension of river networks being either a reactor or an exporter of terrestrial materials is illustrated by the pulse-shunt concept (Raymond et al., 2016), which describes the phenomenon where high-discharge events allow moments for quick pulse/shunting of dissolved organic matter (DOM) through headwaters/small rivers and enhanced export to downstream systems. The temporal alternations of rivers between reaction and export are also illustrated by the concept of river network saturation (RNS; Wollheim et al., 2018), where supply of a constituent surpasses its demand (or removal), resulting in a breakthrough of the constituent from river networks and high exports during high-discharge events. Despite the importance of these earlier works, research that nourished these conceptual developments originated predominantly from temporal observations within small temperate watersheds (0.1-230 km²) (Acuña and Tockner, 2010; Casas-Ruiz et al., 2017; Dhillon and Inamdar, 2013; Raymond and Saiers, 2010; Wilson et al., 2013; Yoon & Raymond, 2012) and lacked a spatial perspective across watershed size or climate conditions. Both conceptual frameworks account for downward shifts of within-network uptake in response to higher discharge in larger watersheds (400–200,000 km²) (Raymond et al., 2016; Wollheim et al., 2018), implying a key spatial constraint on watershed export (Bertuzzo et al., 2017). However, this was not illustrated across watersheds of contrasting reaction rates and/or physical conditions. Given that context, we propose that current research lacks a global geographical context, quantifying how reaction versus export varies across watersheds of different sizes or climate regions due to the lack of a common framework unifying these dynamics.

From a perspective of river network modeling, the efficiency of constituent removal at the watershed/river network scale depends on the amount of time a reactive constituent stays within the system before export, here parameterized as surface water residence time (WRT), assuming negligible delays between transport of the constituent and water. Previous work has identified WRT as a key hydrologic and biogeochemical modulator for aquatic system removal efficiency (Casas-Ruiz et al., 2017; Evans et al., 2017; Maavara, Chen, et al., 2020; Vollenweider, 1975), constituent composition and reactivity (Catalán et al., 2016; Evans et al., 2017; Mari et al., 2007), and autotrophic community dynamics (Obertegger et al., 2007). At the mesoscopic scale, temporary hydraulic retention or residence within land/aquatic functional units of the aquatic "macrosystem" (e.g., hyporheic zones or floodplains) has also been suggested to be critical and linked to constituent removal efficiency in these biogeochemical "hotspots" (Harvey et al., 2019; Marzadri et al., 2017; Newcomer et al., 2018; Zarnetske et al., 2011), which allows for a later-stage scaling-up of these processes at larger spatial scales and/or across time. Despite its apparent significance in watershed biogeochemistry and aquatic ecology (Hosen et al., 2021), plus straightforwardness in calculation (Worrall et al., 2014), surface WRTs have rarely been characterized within or across river networks at broad geographical scales. This was partly due to unavailability of global discharge or flow velocity data sets (Lin et al., 2019) that allows for consistent seasonal characterization of surface WRTs over millions of river/stream reaches and across watersheds. While recently modeled for some regional river systems (Brinkerhoff et al., 2021), this lack of a consistent global residence time database, in practice, has impeded a broad-scale perspective of cross-region watershed removal or export, given that WRT is a necessary parameter.

Constituent removal efficiency depends further on reaction rate, which is often a function of aquatic constituent reactivity, microbial community, kinetic responses to concentration, temperature, and other factors. For a specific reaction and assuming no other significant limitations (e.g., by light or nutrients; Bernhardt et al., 2018), microbial production and respiration are affected by temperature with rates of these processes increasing by a factor of 2–3 for every 10°C increase in temperature (Follstad Shah et al., 2017; Mao & Li, 2018; Peierls & Paerl, 2010; Raymond & Bauer, 2000; Wickland et al., 2012). Though temperature-induced variations in aquatic reaction rate are widespread and important (Follstad Shah et al., 2017; Yvon-Durocher et al., 2010), how they interact with hydrology and the broadly varying WRTs of river networks is largely unknown in terms of shaping the landscape of watershed biogeochemistry and removal/export across climate regions and seasons. Temperature must also be accounted for when determining aquatic ecosystem responses to climate change (Yvon-Durocher et al., 2010, 2011).

Reaction rate in a river is measured either as a volumetric or areal rate (mass length⁻³ (or length⁻²) time⁻¹), an uptake velocity (length time⁻¹), turnover length (length), or simply a per time constant (time⁻¹) (Catalán et al., 2016; Ensign & Doyle, 2006; Mineau et al., 2016; Stream Solute Workshop, 1990). Conceptually and mathematically, all reaction rates can be transformed to a characteristic reaction timescale that corresponds to a specific system-wise removal efficiency (e.g., $t_{1/2}$ or half turnover time that corresponds to a removal efficiency of 1/2) after accounting for the dimension or standing stock of the constituent in the system (Catalán et al., 2016; Marzadri et al., 2017; Zarnetske et al., 2012). This timescale, coupled with the characteristic advection timescale of the reaction system (i.e., WRT), provides a common, unitless indicator (i.e., Damköhler number or Da) based on which the relative reaction intensity in river systems can be evaluated.

In this analysis, we adopted the concept of Da as a common framework and evaluated the spatial and temporal variations of constituent removal in river systems worldwide. The concept was first introduced in the early twentieth century in chemical engineering to describe reaction efficiencies of flow-reaction systems (Rehage & Kind, 2021). We used dissolved organic carbon (DOC) as an illustrative reactive constituent, considering its core role in river biogeochemistry and ecology (Kaplan & Cory, 2016). Depending on the angles of approach, the Da can have different formulations in aquatic studies (Marzadri et al., 2021; Wollheim et al., 2018). In cases where both the advection and reaction timescales are known, the Da is essentially expressed as the advection-to-reaction timescale ratio of the chosen reaction system (unitless, Equation 3 in Section 2). A Da above 1 often refers to a reaction-dominated system, whereas below 1 an advection-dominated system (Rehage & Kind, 2021). Another feature of the Da is its direct translation to a reaction efficiency, given known reaction kinetics (e.g., zero, first, or second order). For instance, under the assumption of first-order kinetics (i.e., reaction rate responds linearly to reactant concentration), the reaction efficiency for reactant to pass through a system can be described as an exponential relationship with Da (i.e., $R = 1 - \exp(-Da)$) (Rehage & Kind, 2021; Wollheim et al., 2018).

The best use of the Da depends further on reaction systems with clearly defined boundaries that allow an advection timescale to be calculated (Marzadri et al., 2017). In this analysis, Da was calculated as a watershed/river network property instead of a reaction site or storage zone property as in many earlier studies that involved Da calculations (Harvey et al., 2019; Zarnetske et al., 2012) to provide a perspective on how the watershed/river network as a whole varies in reaction/export across space and time. We accounted for hydrology-induced spatial and temporal variations in river network advective transport and the temperature effect in the calculation of the Da. Applying the same concept, we also examined how river damming and global warming affect watershed DOC uptake considering their impacts on either WRT (Vörösmarty et al., 1997) or uptake rate (Maavara et al., 2017) in river systems. Through this practice, we aim to provide a unified view for and demonstrate factors that shape the global landscape of river networks as reactors/exporters of reactive terrestrial materials.

2. Materials and Methods

2.1. Instream Water Residence Time

In this analysis, both estimates for instream WRTs and river network Da for DOC uptake were based on the GRADES (Global Reach-Level A Priori Discharge Estimates for SWOT) river networks (Lin et al., 2019). GRADES river networks start channelization at an area threshold of 25 km². A total of ~2.9 million individual flow lines and nine stream orders (SO) (Strahler) were depicted for the global river networks. To delineate watersheds for the GRADES river networks, we utilized topological relationships describing connectivity of all GRADES flow lines. Watersheds, defined as those that export to oceans, were delineated by first locating the outlet (i.e., a flow line with no downstream reaches) and then identifying all upstream flow lines and associated unit catchment areas. The unit catchment areas were then fused to form a single watershed (Python Rasterio, version 1.1). A total of 59,958 watersheds were delineated, ranging in area from 10 to >3 million km². These watersheds form the basis for the current analysis.

GRADES provides reconstructed daily discharge estimates (year 1979–2014) for each river reach (Lin et al., 2019). We used annual and monthly mean discharges from the GRADES to estimate WRTs corresponding to these timescales (i.e., monthly or annual). WRT at a single river reach was calculated as length (m) divided by flow velocity (m s⁻¹) at the river reach. Flow velocity was estimated using a hydraulic geometry formulation of the Manning's equation after imposing a rectangular river channel (Equation 1; Dingman, 2007),

$$V = \left(\frac{S^{0.3}}{n^{0.6} W_b^{0.4}}\right) Q^{0.4} \tag{1}$$

where V is flow velocity (m s⁻¹), S channel slope (unitless), W_b (m) bankfull width of the reach, and Q (m³ s⁻¹) annual/monthly discharge from GRADES. Bankfull widths for the GRADES river networks were from Lin et al. (2020). A uniform Manning's n (0.03) was assumed.

WRT at the river network scale was estimated by routing reach-scale annual/monthly WRTs through the river network. We applied a discharge-weighted algorithm for the routing where cumulative WRTs from all joining upstream reaches were weighted by their respective reach discharge, plus the independently estimated (i.e., reach length divided by flow velocity) advection time at the downstream reach, to obtain an average, cumulative WRT at the downstream reach (Equation 2; Hosen et al., 2021).

$$t_{ri} = \frac{\sum Q_j t_{rj}}{\sum Q_j} + t_i \tag{2}$$

where t_{ri} and t_{rj} (hr) were cumulative WRTs at the downstream reach *i* and the *j*th joining reach, respectively, Q_j water discharge at the *j*th joining reach (m³ s⁻¹), and t_i the advection time at the single downstream reach *i*.

We joined 6,767 dams from the HydroLakes database (Messager et al., 2016) to the GRADES river networks. This dam database, which is the same as GRanD (the Global Reservoir and Dam database) (Lehner et al., 2011), is complete for dams with reservoirs 10 km² or larger, and additionally includes reservoirs to 0.1 km² in area. These dams were distributed in a total of 1,295 watersheds. The HydroLakes database provides annual WRT estimates for each included single reservoir (calculated from statistically modeled reservoir volumes and outflow discharge). To estimate reservoir contribution to river network WRTs at the annual timescale, we replaced WRT at natural GRADES river reaches where HydroLakes reservoirs are situated with HydroLakes reservoir residence times for the river network scale routing. Reservoir wRT was calculated as the difference between the routed WRT with reservoir contribution and without. Reservoir contribution to river network WRTs were available from HydroLakes (Messager et al., 2016).

2.2. River Network Damköhler Number Calculation

In this analysis, monthly Da for DOC uptake was calculated for each of the 59,958 watersheds as the ratio between monthly river network WRTs (t_r , hr) and a temperature-dependent characteristic reaction time for aquatic DOC uptake (t_s , hr) in the system.

$$Da = \frac{t_r}{t_s} \tag{3}$$

The characteristic reaction time (t_s) quantifies the time needed for DOC uptake in a river system (Rehage & Kind, 2021). A Da above 1 means that the system is reaction dominated, whereas a Da below 1 means the system is export dominated. Depending on the understandings of whether riverbed or water column processes dominate DOC uptake in river systems, t_s can be formulated differently (Marzadri et al., 2021). In cases where riverbed processes dominate the uptake (e.g., in small rivers/streams), reaction rate is often expressed as an uptake rate per unit area of stream bottom (U, in unit of mass area⁻¹ time⁻¹) (Stream Solute Workshop, 1990; Wollheim et al., 2006). Uptake velocity (v_f) is defined consequently as the concentration (C, in unit of mass volume⁻¹) normalized areal uptake rate (i.e., $v_f = U/C$). Uptake velocity (v_f) has a unit of length per time and can be thought of as a vertical velocity at which solute (e.g., a reactive DOC molecule) is transferred through the water column to a reactive sediment/water interface (Stream Solute Workshop, 1990). The ratio between water column depth (d, m) and v_f therefore defines a characteristic time needed for the transfer (i.e., $t_s = d/v_f$). The Da for a river system can thus be approximated by the following equation.

$$Da = \frac{t_r/24}{d/v_f} \tag{4}$$

where *d* (m) is a characteristic water column depth for the watershed and v_f (m d⁻¹) is the DOC uptake velocity. For *d*, we used discharge-weighted mean reach water column depth as an approximation to the mean watershed water column depth (Equation 5). By accounting for all river reaches, this formulation accounted for shallower depths (thus more efficient uptake) (Wollheim et al., 2006) at numerous small river reaches and allowed *d* to increase as watershed increases in size (or larger watersheds have deeper *d*) (Bertuzzo et al., 2017).

$$=\frac{\sum Q_i d_i}{\sum Q_i} \tag{5}$$

where Q_i and d_i were discharge and water column depth at the *i*th river reach within a river network, respectively.

d

To determine a t_s or v_f suitable to scale DOC uptake at the watershed scale is difficult, considering the large methodological difficulties in measuring DOC uptake in actual river networks (Fellman et al., 2009; Lutz et al., 2012) and the vast spatial and temporal variations in DOC v_f both within and across river systems (Bott & Newbold, 2013; Mineau et al., 2016; Munn & Meyer, 1990). In practice, constrained by complex and irreplicable molecular compositions, DOM uptake at the reach scale is often evaluated via whole-system simple molecule (e.g., acetate, glucose, and arabinose) or organic leachate addition experiments where added surrogate solute concentration is monitored at multiple distances down experimental reaches of small rivers (Stream Solute Workshop, 1990). Calculated DOC spiraling characteristics from solute addition experiments (median v_f : 3.2 m d⁻¹) however suggest solute elimination within short downstream advection distances (0.9 quantiles: 0.01–1.7 km)

(Table S1 in Supporting Information S1). We suggest that simple molecules or organic leachates represent a highly reactive fraction of the DOM reactivity continuum and overestimate actual DOC uptake in river networks (Mineau et al., 2016). DOC v_f derived from solute addition experiments is thus not suitable to scale DOC uptake at actual watershed scales (Lv et al., 2019).

In this analysis, we have chosen a v_f (0.038 m d⁻¹) derived from a best model fit to hierarchically sampled DOC concentrations along a medium-sized (400 km²) river network (the Ipswich River, USA; Wollheim et al., 2015) to scale Da for instream DOC uptake in global river networks. This v_f corresponds to a balance between total river network DOC gains (e.g., via terrestrial input or autochthonous production) and losses (e.g., via photo-/microbial oxidation) and thus measures net DOC removal within the river network (Wollheim et al., 2015). Importantly, the v_f yields a global DOC removal efficiency that matches the magnitude of global river network CO₂ emission from heterotrophic respiration (Battin et al., 2008; Raymond et al., 2013) (see Section 4). We recognize large uncertainty in applying a single temperature-dependent v_f to whole river network DOC cycling (e.g., inability to account for spatial and temporal variations in aquatic DOC reaction rate; see Section 4) (Bertuzzo et al., 2017; Lv et al., 2019; Raymond et al., 2016; Wollheim et al., 2018). We argue that since the aim of the analysis is to illustrate the impacts of watershed hydrologic conditions and temperature-dependent instream microbial processing on spatial and temporal variations in river network DOC uptake, this v_f offers sufficiently accurate DOC uptake for Da calculation at the watershed scale.

To account for the effects of temperature on uptake and Da, the v_f was assumed to work at an ambient water temperature of 15°C and was scaled up with temperature according to the Arrhenius law (Equation 6; Montagnes et al., 2008).

$$v_{f,T} = v_f \theta^{\frac{T-15^{\circ}C}{10^{\circ}C}}$$
(6)

where $v_{f,T}$ is the uptake velocity at the temperature *T*. An Arrhenius coefficient (θ) of 2 was used in this analysis (Follstad Shah et al., 2017; Mao & Li, 2018; Wickland et al., 2012).

Monthly air temperature for watersheds was estimated by overlapping the GRADES watersheds with a global land surface air temperature data set (World Climate Version 2.0) (Fick & Hijmans, 2017). The monthly air temperature was then converted to water temperature using a simple equation from Raymond et al. (2013) (Equation 7). The relationship derived from an analysis of coupled water and air temperatures from the GLORICH database (Hartmann et al., 2014) and R^2 for the relationship was 0.65 (Raymond et al., 2013).

$$\Gamma_w = 0.67T_a + 7.45 \tag{7}$$

where T_w and T_a are water and air temperatures (°C), respectively. Temperature was bottomed at zero degree Celsius for water. A test against direct measurements at 366 river gauging stations worldwide (United Nations Environment Programme, 2019) shows that the simple calculation predicted sufficiently reliable monthly water temperatures for global watersheds (Figure S1 in Supporting Information S1, $R^2 = 0.8-0.93$).

3. Results

3.1. Surface Water Residence Time

Surface WRT ranges from less than 1 hr to more than 3 months (2,232 hr) for global river networks (with a watershed area of $10->10^6$ km²) (Figure 1a). WRT scales closely with watershed size (Figure 2a). The longest WRTs are found in large watersheds, draining interiors of major continents and shorter WRTs in small watersheds, draining coastal areas or isolated islands (Figure 1a). In terms of SO, median WRT for SO 1 watersheds is 6 hr, about 1 day (23 hr) for SO 2 watersheds, about 10 days (230 hr) for SO 5 watersheds, and over a month (760 hr) for watersheds of SO 7 and above (Figure 1b and Table 1). Using discharge at watersheds' outlet as a weight, a mean "age" for the total exported water from global watersheds could be estimated. This age, corresponding to the mean residence time of surface waters in global river networks, is 908 hr (or 38 days), close to those of SO 7 and 8 watersheds (767–1,396 hr), suggesting the importance of the world's large watersheds in determining the mean surface WRT of land-exported waters.

In addition to system size, WRT also scales negatively with water runoff (or specific discharge) across watersheds (Figure 2c). For instance, for SO 5 watersheds (0.95 quantile drainage area: 2,440–38,000 km²), median WRT in





Figure 1. Maps showing water residence time in (a) global river networks and (b) reservoirs and (c) Damköhler number calculated at mean annual discharge and water temperature.





Figure 2. Surface water residence times in global river networks and their relationship with (a) watershed size, (b) stream order, and (c) water runoff from watersheds.

watersheds of the highest runoff class (>760 mm yr⁻¹) is only 68% that of the lowest runoff class (<80 mm yr⁻¹) (176 vs. 260 hr). The effect of watershed runoff is also apparent in a direct comparison of the largest watersheds (Table S2 in Supporting Information S1). As an illustration, the Amazon has the largest watershed area (over 3 million km², 1,730 hr), but the longest WRT is found in the Yellow River, a semiarid river with a watershed area of only 16% that of the Amazon (0.48 million km², 2,160 hr). In addition, while the Yellow and Mekong are comparable in watershed size (0.48 vs. 0.42 million km²), WRT in the Mekong (which flows partly through tropical Southeast Asia) is only 75% of the Yellow River (1,670 vs. 2,160 hr).

3.2. Damköhler Number

Da calculated for river network DOC uptake spans over five orders of magnitude (from <0.001 to >90; the 0.90 quantiles are 0.01–3.2; Figure 1c), suggesting highly variable river network DOC uptake across watersheds. The smallest Da (e.g., <0.5; and thus export-oriented systems) is found in small river networks (SO 1–3) (Figures 3a and 4), draining coastal watersheds or off-continent islands (Figure 1c). These systems make up the largest percentage among the global watersheds (\sim 93%) but a rather small fraction (\sim 12.3%) of the total global water discharge (~47,000 km³ yr⁻¹) (Lin et al., 2019) because of their small drainage size (e.g., ~95% are <1,000 km²) (Table 1). These watersheds are also found at all latitudes (Figure 1c) despite the large temperature gradient along the altitude (e.g., mean annual air temperature from $<5^{\circ}$ C in the high latitudes to $>25^{\circ}$ C in the tropics).

The Da for more than half of SO 7–9 watersheds is larger than 2, suggesting uptake dominance in these largest river networks. These watersheds make up 46% of the global land surface area, and summed discharge from these watersheds (~25,100 km³ yr⁻¹) is more than half (54%) of the total global water discharge (Table 1). In terms of

Table 1
Global Watershed Area, Annual Discharge, Stream Network Length, and Annual Water Residence Times by Watershed
Stream Order

Stream	No. of	Watershed area	Annual	Stream network length (km)	Annual wate	r residence time (hr)
order	watersheds	(km ²)	discharge (km ³)		Mean \pm SD	Median (range)
1	32,616	$66 \pm 1,128$	0.04 ± 0.1	9 ± 10	10 ± 12	6 (0.03–386)
2	15,658	$299 \pm 6{,}405$	0.1 ± 0.2	48 ± 40	30 ± 25	23 (0.4–449)
3	7,365	$942 \pm 7,399$	0.4 ± 0.8	205 ± 152	68 ± 48	55 (6-627)
4	2,943	$3,827 \pm 33,493$	1 ± 3 .	839 ± 608	137 ± 84	117 (10-822)
5	1,009	$15,915 \pm 117,353$	6 ± 12	$3,602 \pm 2,715$	258 ± 133	227 (47–1,157)
6	276	$49,375 \pm 42,073$	22 ± 37	$15,753 \pm 13,473$	468 ± 229	411 (176–1,431)
7	64	$219,103 \pm 185,081$	75 ± 121	$61,302 \pm 45,577$	805 ± 338	767 (322–2,233)
8	23	$1,041,948 \pm 719,893$	419 ± 349	$296{,}721 \pm 160{,}466$	$1,\!435\pm395$	1,396 (761–2,191)
9	4	$2,348,999 \pm 530,711$	$2,\!658\pm2,\!861$	$752,162 \pm 254,567$	$1,\!782\pm252$	1,697 (1,584–2,147)



Global Biogeochemical Cycles



Figure 3. Damköhler number (Da) for dissolved organic carbon uptake in global river networks by (a) stream order, (b) watershed runoff, and (c) latitudinal. Shaded areas show total water discharge from watersheds of each subgroup. Dashed line shows Da = 1.



Figure 4. Monthly Damköhler number for dissolved organic carbon uptake in global river networks by stream order and climate region. Dashed line shows Da = 1.

SO, Das are 0.15–0.4 for SO 1–3 watersheds, 0.6–1.3 for SO 4–6 watersheds, and 1.8–2.8 for SO 7–9 watersheds (Figure 3a). Da turns above 1 between SO 5 and 6 (or 9,300–38,000 km² in watershed area), implying that watersheds turn from export-to reaction-dominated systems in terms of DOC uptake at the same SO. Despite large seasonal variations, large watersheds (e.g., SO 7–9) in temperate and tropical regions are primarily reaction-dominated systems as suggested by Da above 1 in all seasons (Figure 4).

Besides in large watersheds, high Da (e.g., 3–23) are also found in arid basins of Sahara, central Asia, southwestern US, and central Australia (Figure 1c), suggesting these watersheds are reaction dominated. The Da is 3.25 for the lowest watershed runoff class (<80 mm yr⁻¹), ~2 times that in river systems of higher runoff (i.e., >80 mm yr⁻¹) (Figure 3b). Despite high Da, total water discharge from arid watersheds is only ~1.3% of the global total discharge.

Across climate regions, Das are highest for tropical rivers (absolute latitude of watershed outlet is below 23.5° ; 1.82), intermediate in temperate rivers (23.5° - 56° ; 1.48–1.74), and lowest in the Arctic rivers (above 56° ; 0.61) (Figure 3c). Total water discharge is 58%, 29%, and 13% of the global total for tropical, temperate, and the Arctic watersheds, respectively. In contrast to high Da in temperate and tropical river systems, Da in the Arctic rivers are low (e.g., less than 1) only until the largest watersheds (SO 8) (Figure 4), suggesting export dominance in most Arctic rivers.

3.3. Damming Effect

For a total of 1,295 watersheds found to be associated with at least one reservoir, WRT ranges from 6 hr to >53 years in comparison to 1 hr to 93 days under natural flows in these watersheds (Figures 1b and 6a). Damming thus substantially extends the time that water stays in these watersheds by 1–86,800 times. By SO, WRTs in the regulated watersheds are 4.7–823 times as long as under natural flows and the changes are most significant in small watersheds (e.g., 125–823 times in SO 1–3 watersheds) (Figure 6b). This translates to an enhancement in Da of 141–2,944 times from 0.08 to 2.7 under natural flows to 13–232 under dammed conditions (Figure 6c). Damming therefore transforms natural rivers to systems of high processing capacities, assuming the same substrate uptake rate in reservoirs as in natural rivers (Wollheim et al., 2015). Along the latitude, enhancement in Da by river damming is most significant in northern and southern temperate watersheds where most reservoirs are found (Figure 5c).

3.4. Global Warming Effect

Evaluating on an annual basis, over 98% of the global watersheds had seen an increase in temperature (averaging 0.63° C) between the year 1900 and 2006, which causes an increase in river network Da for DOC uptake of ~4.9% in magnitude at the global scale (Figure 7a). Under future climate changes, all watersheds will experience temperature increase and the estimated Da increases by the year 2100 are 19% and 42% under the RCP4.5 and RCP8.5 scenarios, respectively (Figures 7b and 7c). In line with the projected temperature changes, Da increases are more pronounced in the Arctic and northern temperate rivers. For example, Da increases under the RCP4.5 scenario are 18%, 18%, 10%, and 7% for the Arctic, northern temperate, tropical, and southern temperate rivers, respectively, and 55%, 34%, 23%, and 17%, respectively, under the RCP8.5 scenario. These Da increases imply enhanced instream DOC uptake in global river networks under future climate changes.

4. Discussion

4.1. DOC Cycling and Measurements in River Networks

DOC cycling in river networks involves multiple processes that contribute simultaneously to its gains/losses within rivers (Battin et al., 2008; Bernhardt et al., 2018; Kaplan & Cory, 2016). In situ photooxidation (Maavara et al., 2021) and biotic mineralization (del Giorgio & Pace, 2008; Raymond & Bauer, 2000) are the two main processes contributing to permanent DOC removal and CO_2 emission from river systems. The magnitude of DOC removal rate in river networks is however difficult to quantify due to the complex interplays among multiple biotic/abiotic processes that contribute to DOC concentration variations in spatially distributed river networks (Lv et al., 2019; Wollheim et al., 2015). Characterization of river network DOC removal is also hindered by complex DOM compositions that preclude replication and realistic DOC uptake rate quantification via simple



Global Biogeochemical Cycles



Figure 5. Latitudinal variations in Damköhler number for dissolved organic carbon uptake in (a) natural river networks and (b) river networks with the damming effect added. Shaded areas show latitudinal variations in water discharge from global watersheds.

solute addition experiments at river reaches (Mineau et al., 2016; Stream Solute Workshop, 1990). Bottle river/ stream water photooxidation and/or microbial incubation experiments (i.e., bioassays) provide opportunities for bulk DOC reactivity quantification in an isolated and controlled setting (del Giorgio & Pace, 2008). Bioassays are however problematic because the more important benthic/hyporheic removals in real river network settings are not accounted for (Harvey et al., 2019; Newcomer et al., 2018).

In Table S1 in Supporting Information S1 and Figure 8, we compared DOC spiraling characteristics measured from reach-scale solute addition experiments and bulk photooxidation/bioassay experiments. Median DOC v_f measured from 152 individual simple molecule or organic leachate addition experiments was 3.2 m d⁻¹ (0.9 quantiles: 0.9–22 m d⁻¹). The v_f suggests a characteristic reaction time (t_r) of ~0.3 hr (0.9 quantiles: 0.03–9.7 hr) (Figure 8) or a characteristic uptake length of 3.7 km (0.9 quantiles: 0.27–40 km) (Table S1 in Supporting Information S1). These DOC spiraling characteristics suggest quick elimination of the added surrogate solutes within water advection timescales even of the smallest watersheds (e.g., 6 hr in SO 1 watersheds, Table 1). Mineau et al. (2016) tried to reconcile the solute addition-derived stream DOC v_f by accounting for DOC reactivity under ambient conditions. The resultant DOC v_f (0.37 m d⁻¹) was one order of magnitude lower than measured from stream solute addition experiments. However, runoff DOC concentration required to balance the removal rate was high (14.8 in comparison to 4.3 mg L⁻¹ in typical headwater streams) (Mineau et al., 2016), suggesting that the ambient DOC v_f may still represent an overestimate. We suggest simple molecules or organic leachates





Figure 6. Annual water residence time (a, b) and mean Damköhler number (c) for dissolved organic carbon uptake in reservoir-affected watersheds.

represent a highly labile fraction and rapidly consumed within the heterogeneous aquatic DOM mixtures (Lutz et al., 2012). In addition, high-solute uptake can also result from assimilatory biotic uptake, transformation, or adsorption (Munn & Meyer, 1990) and does not correspond solely to DOC mineralization.

In contrast to rapid removals suggested from solute addition experiments, median DOC t_s calculated from 90 dark bioassays (as the inverse of first-order decay coefficient or k) was five orders of magnitude higher than that measured from reach solute addition experiments (11,100 hr vs. 0.3 hr) (Figure 8). The bioassay t_s was also fivefold the longest WRT found for river networks (2,232 hr, see Section 3). The bioassay t_s thus implies limited DOC reactivity or removal even in the largest river networks (Da <0.2), probably because the method significantly underestimates in situ DOC uptake by accounting for processing only within water columns (del Giorgio & Pace, 2008). DOC t_s measured from 20 combined photooxidation/bioassays were comparatively much lower (median: 314 hr) (Figure 8), indicating quicker mineralization when considering in situ photo- and biotic oxidations together. Particularly, photodegradation is a widely recognized process that promotes microbial breakdown (Allesson et al., 2016; Lu et al., 2013) though either photo- or microbial oxidation alone is insignificant for DOC removal in river networks (del Giorgio & Pace, 2008; Maavara et al., 2021). DOC v_f chosen for Da scaling in this analysis (0.038 m d⁻¹, corresponding of a t_s of 720 hr or WRT of a 6–7 order river network) (Wollheim et al., 2015) falls between DOC uptake determined from the solute addition and dark bioassay methods (Figure 8). Slightly slower turnover in comparison to that suggested from photooxidation/bioassays (720 hr vs. 314 hr) results likely from the fact that the model-derived DOC spiraling characteristics (v_f or t_s) (Wollheim et al., 2015) accounted for in situ autochthonous DOC production in real river networks.

The large discrepancy in DOC uptake measured from different methods highlights incompleteness in current understandings of DOC cycling in river networks (Bernhardt et al., 2018; Raymond et al., 2016). Nonetheless, our choice of v_f yields reasonable instream DOC removal at the global scale. Under this v_f , global river network DOC removal efficiency was ~65% (as a discharge-weighted mean removal efficiency across global watershed; corresponding to a Da of 1.05 under the first-order reaction kinetics). Considering total DOC export by global river networks is ~0.21 Pg C yr⁻¹ (Ludwig et al., 1996), total instream respiratory DOC loss would be at a magnitude of ~0.6 Pg C yr⁻¹. This value corresponds to a third of total CO₂ emitted from global river networks (1.8–2.0 Pg C yr⁻¹) (Liu et al., 2022; Raymond et al., 2013) and within the empirical range of estimated DOC metabolism contribution to CO₂ emission in typical river systems (15%–50%) (Battin et al., 2008; Hotchkiss et al., 2015; Ward et al., 2013). We however recognize rarity of system-wise DOC removal estimates (Wollheim et al., 2015) and the limitations in describing the complex issue of river network DOC cycling as a single v_f or Da (see Section 4.5). The work is however valuable as it provides a means to examine the variability of DOC removal versus export across watersheds at broad geographical scales.

4.2. Watershed Controls on DOC Removal and Export

The analysis identifies watershed size to be a primary factor determining the relative balance between DOC removal and export in river networks. This effect is best illustrated by a comparison of Da in river networks at the two ends of the watershed size continuum (Figures 3 and 4). Drainage area of the largest watersheds (i.e., SO 7 and plus) is typically >3,000-50,000 times as large as the smallest watersheds (e.g., SO 1), which results in a difference in river network WRT of typically 100–300 times and a difference in Da of 12–20 times. The large difference





Figure 7. Historical (a) and projected (b, c) Damköhler number changes for dissolved organic carbon uptake in global watersheds. Historical Damköhler number change is calculated based on the temperature difference between the average of the years 1900–1910 and the years of 2006–2016.





Figure 8. Characteristic reaction timescales for dissolved organic carbon (DOC) uptake (t_s) determined from simple molecule or leachate addition experiments, bioassay/photooxidation experiments, and a river network DOC cycling modeling work (Wollheim et al., 2015). Data are from published literature (see Table S1 in Supporting Information S1). For conversion methods from original reported literature data, see annotations to Table S1 in Supporting Information S1. A high t_s indicates low uptake rate.

in watershed size and river network WRT thus imposes a physical constraint on DOC removal in the smallest versus largest watersheds. For instance, a 20°C temperature difference (i.e., close to the largest seasonal or latitudinal water temperature difference) would cause a fourfold change in Da (Equation 6). This change however wound not make low-ordered stream networks (with Da <0.25, making up 66% of SO 1–3 river networks) reaction-oriented systems. This physical constraint therefore determines the smallest river networks (SO 1–3) to be predominantly export-oriented systems (Da << 1) in line with observations that suggest effective river organic carbon exports by small mountainous watersheds (Hilton et al., 2011) and the largest river networks (SO 7–9) reaction-oriented systems (Da >1) across seasons and the latitude (Figures 1c and 4) though latitudinal and seasonal variations within each SO are also significant (Figure 4).

The turn from export-to reaction-oriented systems at ca. SO 5–6 for global watersheds (Figures 3 and 4) is important as it implies a clear difference in the amount and quality of DOC exported by river networks of different sizes to receiving basins. While small watersheds principally export terrestrial DOC in less altered forms, large watersheds export highly processed DOC that preserves less terrestrial compositional signals (Hosen et al., 2021). The analysis further suggests that this turn can occur at variable watersheds become reaction-oriented at a smaller watershed size than wetter and colder watersheds (Figure 4).

On temporal scales, high-discharge events are suggested to induce large exports due to enhanced DOC concentration and shortened WRT that limit within-network removal (Raymond et al., 2016; Yoon & Raymond, 2012). The analysis suggests this phenomenon is also subject to the watershed size effect. For instance, a high-discharge event of 10 times as large as the base flow would increase flow velocity by a factor of ~2.5 (Equation 1) and decrease instream WRT correspondingly by ~60%. Therefore, assuming the same reaction rate (or reaction timescale t_r , Equation 3) for DOC, river networks with Da between 1 and 2.5 are most prone to the high-discharge effect. This corresponds roughly to medium-sized river networks (SO 5–7), which alternate between reactionand export-oriented systems with flow and seasonal temperature changes (Figure 4). In contrast, the smallest river networks remain export-oriented and the largest river networks reaction-oriented under high flows.

In large watersheds where high-discharge-induced DOC transport does not end up being exported, hot spots for DOC uptake perpetuate to downstream ecosystems within the river networks. The progressive saturation in river network uptake capacity and quick transfers of DOC through headwaters and small rivers under high discharge and increased loads have been described as the "shunting" effect within the pulse-shunt framework (Raymond et al., 2016) or "spatial saturation" under the RNS concept (Wollheim et al., 2018). These concepts underscore the capacity for large river networks to process legacy constituents leaked from upstream river networks, a capacity that small river networks do not possess.

4.3. Latitudinal and Seasonal Changes

The temperature effect causes a DOC uptake gradient along the latitude (Figures 3–5). The analysis suggests high DOC removal capacity in tropical rivers, which is overall two times as large as that in temperate rivers and 4–6 times as large as that in high-latitude rivers in terms of Da (Figures 3–5). Therefore, enhanced DOC uptake due to higher temperatures is more pronounced though higher discharge in tropical river networks works to shorten instream residence times and reduce removal. High DOC removal is in line with quick DOC turnover (Mayorga et al., 2005) and large CO₂ emission (Richey et al., 2002) in tropical river networks. In contrast, the Arctic rivers are identified as export-oriented systems where main export dominance can be found in systems of up to the 7th SO (Figures 3 and 4).

Temperature-induced Da change across seasons is more pronounced in medium-sized to large (SO 6–9) river networks in temperate and the Arctic regions (Figures 4 and 5). We propose that temperate and the Arctic rivers

are characterized by much greater seasonal variations in temperature and DOC uptake rate than tropical rivers (e.g., 0°C–23°C vs. 23°C–25°C), which cause larger seasonal Da changes than in tropical rivers. Additionally, longer WRTs in large watersheds allow for more sufficient DOC processing (Lv et al., 2019; Wollheim et al., 2018) and have an enlarged and more pronounced temperature effect in large (SO 6–9) than in smaller river systems. Variations in Da among geographical regions are also observable. For example, SO 6 watersheds in southern temperate regions exhibit significantly higher Da than in other climatic regions (Figures 4 and 5). We suggest that a larger percentage of arid/semiarid watersheds in this area (e.g., in central/western Australia) is responsible for the elevated Da, considering that arid/semiarid river systems also feature longer retention times and thus higher processing capacity (Figure 1). This reinforces the idea of a combined effect of hydrology and aquatic biogeochemistry in shaping the riverine DOC cycling landscapes across global watersheds.

4.4. The Impact of River Damming and Global Warming

We note the significant impacts of river impounding and climate warming on WRT and/or DOC uptake in river networks (Figures 6 and 7). Dam-induced water residence extension in river networks has also been described by the concept of river water "aging" (Vörösmarty et al., 1997). The magnitude of damming-induced river water aging depends highly on the number and spatial distribution of constructed dams and therefore the completeness of dam database matters. Evaluating based on a total of 746 large reservoirs (capacity >0.5 km³), Vörösmarty et al. (1997) arrived at a mean river water aging of 31 days (or 744 hr) for global watersheds or 58 days (1.392 hr) for regulated watersheds alone. Here, estimating based on the HydroLakes database that includes all reservoirs >10 km² in area and additional reservoirs to 0.1 km² (Messager et al., 2016), we suggest a mean river water aging by dams of 237 days (5,687 hr) for global watersheds or 371 days (8,908 hr) for regulated watersheds alone. Therefore, river damming has increased water aging by approximately seven times since its first evaluation ~24 years ago (Vörösmarty et al., 1997). More significant water aging and reservoir impacts on river network DOC uptake are anticipated with the ongoing and planned dam constructions on the world's rivers (Zarfl et al., 2015). We acknowledge systematic regime shifts associated with these changes and under climate change (Maavara, Chen, et al., 2020; Maavara et al., 2017) and future research needs on this.

4.5. Caveats and Limitations

The main limitation of using a single, temperature-dependent v_f or t_s to describe DOC removal is its inability to account for the significant spatial and temporal variability in DOC reactivity and complex DOC cycling dynamics in real river networks (Wollheim et al., 2015). First, aquatic DOC composition is known to vary dramatically in rivers of different sizes, watershed compositions, and geographic regions (Kaplan & Cory, 2016) and cause significant variations in in situ aquatic DOC reactivity. This effect is however difficult to account for considering current technical difficulties in quantifying in situ DOC reactivity (Mineau et al., 2016). Second, terrestrial landscapes are heterogenous in terms of DOC supply to river networks and therefore the within-network location of terrestrial DOC sources (e.g., wetlands) matter in terms of allochthonous DOC input and removal in river networks (Lv et al., 2019; Wollheim et al., 2015). The spatial heterogeneity in terrestrial DOC supply was however not accounted for by discharge-weighted WRT estimates in this analysis (Equation 2), which implicitly assumed spatially uniform input from terrestrial landscapes. Third, rivers of different sizes or hydrogeomorphic configurations are expected to possess different DOC metabolic capabilities (Battin et al., 2008). This variability was not accounted for by a v_f designed to respond solely to temperature in this analysis (Equation 6). Fourth, the single v_f also did not account for within-network variations in autochthonous DOC production, particularly in impounded waters where autochthonous production strengthens and aquatic DOC dynamics alters (Maavara, Chen, et al., 2020).

Despite the limitation, the Da analysis allows us a unified view of riverine DOC reaction variability across broad spatial scales. One main value of the current analysis is the common framework for scaling DOC removal and the hydrologic advection timescales that are necessary for a quantitative assessment of the magnitude of DOC removal in global river networks. We suggest that future river network DOC modeling work takes into consideration the hydrologic and biogeochemical perspectives for quantitative evaluations of DOC cycling in river networks. We however acknowledge that the current analysis lacks a full spatially resolved DOC data set necessary for quantitative DOC budget assessments in global river networks. Efforts are also needed for a full disentangling of the complex DOC reaction rate variations in river networks.

We also note the operational nature in defining river networks as reaction- or export-oriented systems using a conventional Da cutoff of 1 (Rehage & Kind, 2021). This Da corresponds to a high removal efficiency (63%) under the first-order reaction kinetics assumption and suggests a transition from export- to reaction-oriented river networks at around SO 5–6 for DOC uptake. Using 0.1 as the cutoff, which corresponds to a removal efficiency of ~10% instead, only the smallest (e.g., first order) river networks would be classified as export-dominated systems. Furthermore, Da and river network removals are highly sensitive to the magnitude of v_f (Bertuzzo et al., 2017; Lv et al., 2019; Wollheim et al., 2018). Therefore, export-dominated river networks defined in terms of DOC uptake may be highly reactive systems in terms of the cycling of other constituents with orders of magnitude higher uptake (e.g., nitrogen or phosphorous) (Wollheim et al., 2018). Therefore, the DOC-based classification of reaction- or export-oriented river networks should be interpreted in a relative sense and within the context of a targeted constituent.

5. Conclusion

The concept of Da relates the reaction to advection timescales of reactive constituent transport and provides a common framework for evaluation of river network removal/export across space and time. In this analysis, we viewed watersheds/river networks as unit reaction systems and evaluated factors that affect their reaction efficiency across watersheds and seasons. We demonstrate that the watershed size imposes a primary control on river network reaction due to a difference in WRT of >3 orders of magnitude. Hydrology or temperature-induced seasonal variations in reaction (or export) are more pronounced in medium-sized watersheds, suggesting that these watersheds are more transitional between reactors and exporters across seasons or hydrologic variations. Temperature-induced latitudinal variations in reaction are also significant, where tropical river systems are two times more efficient than temperate river systems and 4–6 times more efficient than the Arctic river systems. Assessing under the same framework, river damming has a profound effect on river network reaction, considering significantly extended WRTs especially for small watersheds. Global warming is projected to increase reaction by ca. 19% until year 2100 under the RCP4.5 scenario. We conclude that river network reaction/export is controlled together by watershed size, constituent reactivity, and spatiotemporal variations in hydrology and temperature.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The GRADES (Global Reach-Level A Priori Discharge Estimates for SWOT) river networks and daily discharge data sets are accessible via https://www.reachhydro.org/home/records/grades. Monthly water residence time and Damköhler number estimates for DOC cycling in global watersheds are available via Dryad at https://doi.org/10.5061/dryad.2v6wwpzqv.

References

Acuña, V., & Tockner, K. (2010). The effects of alterations in temperature and flow regime on organic carbon dynamics in Mediterranean river networks. *Global Change Biology*, 16(9), 2638–2650. https://doi.org/10.1111/j.1365-2486.2010.02170.x

Allesson, L., Strom, L., & Berggren, M. (2016). Impact of photochemical processing of DOC on the bacterioplankton respiratory quotient in aquatic ecosystems. *Geophysical Research Letters*, 43(14), 7538–7545. https://doi.org/10.1002/2016gl069621

Battin, T. J., Kaplan, L. A., Findlay, S., Hopkinson, C. S., Marti, E., Packman, A. I., et al. (2008). Biophysical controls on organic carbon fluxes in fluvial networks. *Nature Geoscience*, 1(2), 95–100. https://doi.org/10.1038/ngeo101

Bauer, J. E., Cai, W.-J., Raymond, P. A., Bianchi, T. S., Hopkinson, C. S., & Regnier, P. A. G. (2013). The changing carbon cycle of the coastal ocean. *Nature*, 504(7478), 61–70. https://doi.org/10.1038/nature12857

Bernhardt, E. S., Heffernan, J. B., Grimm, N. B., Stanley, E. H., Harvey, J. W., Arroita, M., et al. (2018). The metabolic regimes of flowing waters. Limnology & Oceanography, 63(S1), S99–S118. https://doi.org/10.1002/lno.10726

Bertuzzo, E., Helton, A. M., Hall, R. O., & Battin, T. J. (2017). Scaling of dissolved organic carbon removal in river networks. Advances in Water Resources, 110, 136–146. https://doi.org/10.1016/j.advwatres.2017.10.009

Bott, T. L., & Newbold, J. D. (2013). Ecosystem metabolism and nutrient uptake in Peruvian headwater streams. International Review of Hydrobiology, 98(3), 117–131. https://doi.org/10.1002/iroh.201201612

Brinkerhoff, C. B., Raymond, P. A., Maavara, T., Ishitsuka, Y., Aho, K. S., & Gleason, C. J. (2021). Lake morphometry and river network controls on evasion of terrestrially sourced headwater CO₂. *Geophysical Research Letters*, 48(1), e2020GL090068. https://doi.org/10.1029/2020gl090068

Acknowledgments

This work was financially supported by the National Key Research and Development Program of China (2021YFC3200401), the Fundamental Research Funds for the Central Universities (2020NTST13), the National Aeronautics and Space Administration Award (NNX17AI74G), and the National Science Foundation Award (1340749).

- Casas-Ruiz, J. P., Catalan, N., Gomez-Gener, L., von Schiller, D., Obrador, B., Kothawala, D. N., et al. (2017). A tale of pipes and reactors: Controls on the in-stream dynamics of dissolved organic matter in rivers. *Limnology & Oceanography*, 62(S1), S85–S94. https://doi. org/10.1002/lno.10471
- Catalán, N., Marce, R., Kothawala, D. N., & Tranvik, L. J. (2016). Organic carbon decomposition rates controlled by water retention time across inland waters. *Nature Geoscience*, 9(7), 501–504. https://doi.org/10.1038/ngeo2720
- Cole, J. J., Prairie, Y., Caraco, N., McDowell, W., Tranvik, L., Striegl, R., et al. (2007). Plumbing the global carbon cycle: Integrating inland waters into the terrestrial carbon budget. *Ecosystems*, 10(1), 172–185. https://doi.org/10.1007/s10021-006-9013-8
- del Giorgio, P. A., & Pace, M. L. (2008). Relative independence of dissolved organic carbon transport and processing in a large temperate river: The Hudson river as both pipe and reactor. *Limnology & Oceanography*, 53(1), 185–197. https://doi.org/10.4319/lo.2008.53.1.0185
 - Dhillon, G. S., & Inamdar, S. (2013). Extreme storms and changes in particulate and dissolved organic carbon in runoff: Entering uncharted waters? *Geophysical Research Letters*, 40(7), 1322–1327. https://doi.org/10.1002/grl.50306
 - Dingman, L. S. (2007). Analytical derivation of at-a-station hydraulic–geometry relations. Journal of Hydrology, 334(1), 17–27. https://doi. org/10.1016/j.jhydrol.2006.09.021
 - Ensign, S. H., & Doyle, M. W. (2006). Nutrient spiraling in streams and river networks. Journal of Geophysical Research, 111(G4). https://doi. org/10.1029/2005jg000114
 - Evans, C. D., Futter, M. N., Moldan, F., Valinia, S., Frogbrook, Z., & Kothawala, D. N. (2017). Variability in organic carbon reactivity across lake residence time and trophic gradients. *Nature Geoscience*, 10(11), 832–835. https://doi.org/10.1038/ngeo3051
 - Fellman, J. B., Hood, E., Edwards, R. T., & Jones, J. B. (2009). Uptake of allochthonous dissolved organic matter from soil and salmon in coastal temperate rainforest streams. *Ecosystems*, 12(5), 747–759. https://doi.org/10.1007/s10021-009-9254-4
- Fick, S. E., & Hijmans, R. J. (2017). WorldClim 2: New 1-km spatial resolution climate surfaces for global land areas. International Journal of Climatology, 37(12), 4302–4315. https://doi.org/10.1002/joc.5086
- Follstad Shah, J. J., Kominoski, J. S., Ardon, M., Dodds, W. K., Gessner, M. O., Griffiths, N. A., et al. (2017). Global synthesis of the temperature sensitivity of leaf litter breakdown in streams and rivers. *Global Change Biology*, 23(8), 3064–3075. https://doi.org/10.1111/gcb.13609
- Hartmann, J., Lauerwald, R., & Moosdorf, N. (2014). A brief overview of the GLObal RIver CHemistry Database, GLORICH. In J. Gaillardet (Ed.), Geochemistry of the Earth's Surface Ges-10 (pp. 23–27). Elsevier Science Bv.
- Harvey, J., Gomez-Velez, J., Schmadel, N., Scott, D., Boyer, E., Alexander, R., et al. (2019). How hydrologic connectivity regulates water quality in river corridors. JAWRA Journal of the American Water Resources Association, 55(2), 369–381. https://doi.org/10.1111/1752-1688.12691
- Hilton, R. G., Galy, A., Hovius, N., Horng, M.-J., & Chen, H. (2011). Efficient transport of fossil organic carbon to the ocean by steep mountain rivers: An orogenic carbon sequestration mechanism. *Geology*, 39(1), 71–74. https://doi.org/10.1130/g31352.1
- Hosen, J. D., Allen, G. H., Amatulli, G., Breitmeyer, S., Cohen, M. J., Crump, B. C., et al. (2021). River network travel time is correlated with dissolved organic matter composition in rivers of the contiguous United States. *Hydrological Processes*, 35(5), e14124. https://doi. org/10.1002/hyp.14124
- Hotchkiss, E. R., Hall, R. O., Jr., Sponseller, R. A., Butman, D., Klaminder, J., Laudon, H., et al. (2015). Sources of and processes controlling CO₂ emissions change with the size of streams and rivers. *Nature Geoscience*, 8(9), 696–699. https://doi.org/10.1038/ngeo2507
- Kaplan, L. A., & Cory, R. M. (2016). Dissolved organic matter in stream ecosystems: Forms, functions, and fluxes of watershed tea (pp. 241–320). Academic Press Ltd-Elsevier Science Ltd.
- Lehner, B., Liermann, C. R., Revenga, C., Vorosmarty, C., Fekete, B., Crouzet, P., et al. (2011). High-resolution mapping of the world's reservoirs and dams for sustainable river-flow management. *Frontiers in Ecology and the Environment*, 9(9), 494–502. https://doi.org/10.1890/100125
- Lin, P., Pan, M., Allen, G. H., de Frasson, R. P., Zeng, Z., Yamazaki, D., & Wood, E. F. (2020). Global estimates of reach-level bankfull river width leveraging big data geospatial analysis. *Geophysical Research Letters*, 47(7), e2019GL086405. https://doi.org/10.1029/2019gl086405
- Lin, P., Pan, M., Beck, H. E., Yang, Y., Yamazaki, D., Frasson, R., et al. (2019). Global reconstruction of naturalized river flows at 2.94 million reaches. *Water Resources Research*, 55(8), 6499–6516. https://doi.org/10.1029/2019wr025287
- Liu, S., Kuhn, C., Amatulli, G., Aho, K., Butman, D. E., Allen, G. H., et al. (2022). The importance of hydrology in routing terrestrial carbon to the atmosphere via global streams and rivers. *Proceedings of the National Academy of Sciences of the United States of America*, 119(11), e2106322119. https://doi.org/10.1073/pnas.2106322119
- Lu, Y., Bauer, J. E., Canuel, E. A., Yamashita, Y., Chambers, R. M., & Jaffé, R. (2013). Photochemical and microbial alteration of dissolved organic matter in temperate headwater streams associated with different land use. *Journal of Geophysical Research: Biogeosciences*, 118(2), 566–580. https://doi.org/10.1002/jgrg.20048
- Ludwig, W., Probst, J.-L., & Kempe, S. (1996). Predicting the oceanic input of organic carbon by continental erosion. *Global Biogeochemical Cycles*, 10(1), 23–41. https://doi.org/10.1029/95gb02925
- Lutz, B. D., Bernhardt, E. S., Roberts, B. J., Cory, R. M., & Mulholland, P. J. (2012). Distinguishing dynamics of dissolved organic matter components in a forested stream using kinetic enrichments. *Limnology & Oceanography*, 57(1), 76–89. https://doi.org/10.4319/lo.2012.57.1.0076
- Lv, S., Yu, Q., Wang, F., Wang, Y., Yan, W., & Li, Y. (2019). A synthetic model to quantify dissolved organic carbon transport in the Changjiang River system: Model structure and spatiotemporal patterns. *Journal of Advances in Modeling Earth Systems*, 11(9), 3024–3041. https://doi. org/10.1029/2019ms001648
- Maavara, T., Akbarzadeh, Z., & Van Cappellen, P. (2020). Global dam-driven changes to riverine N:P:Si ratios delivered to the coastal ocean. Geophysical Research Letters, 47(15), e2020GL088288. https://doi.org/10.1029/2020gl088288
- Maavara, T., Chen, Q., Van Meter, K., Brown, L. E., Zhang, J., Ni, J., & Zarfl, C. (2020). River dam impacts on biogeochemical cycling. Nature Reviews Earth & Environment, 1(2), 103–116. https://doi.org/10.1038/s43017-019-0019-0
- Maavara, T., Lauerwald, R., Regnier, P., & Van Cappellen, P. (2017). Global perturbation of organic carbon cycling by river damming. Nature Communications, 8(1), 15347. https://doi.org/10.1038/ncomms15347
- Maavara, T., Logozzo, L., Stubbins, A., Aho, K., Brinkerhoff, C., Hosen, J., & Raymond, P. (2021). Does photomineralization of dissolved organics matter in temperate rivers? *Journal of Geophysical Research: Biogeosciences*, 126(7), e2021JG006402. https://doi.org/10.1029/2021jg006402
- Mao, R., & Li, S. Y. (2018). Temperature sensitivity of biodegradable dissolved organic carbon increases with elevating humification degree in subtropical rivers. *The Science of the Total Environment*, 635, 1367–1371. https://doi.org/10.1016/j.scitotenv.2018.04.256
- Mari, X., Rochelle-Newall, E., Torréton, J.-P., Pringault, O., Jouon, A., & Migon, C. (2007). Water residence time: A regulatory factor of the DOM to POM transfer efficiency. *Limnology & Oceanography*, 52(2), 808–819. https://doi.org/10.4319/lo.2007.52.2.0808
- Marzadri, A., Amatulli, G., Tonina, D., Bellin, A., Shen, L. Q., Allen, G. H., & Raymond, P. A. (2021). Global riverine nitrous oxide emissions: The role of small streams and large rivers. *The Science of the Total Environment*, 776, 145148. https://doi.org/10.1016/j.scitotenv.2021.145148
- Marzadri, A., Dee, M. M., Tonina, D., Bellin, A., & Tank, J. L. (2017). Role of surface and subsurface processes in scaling N₂O emissions along riverine networks. *Proceedings of the National Academy of Sciences of the United States of America*, 114(17), 4330–4335. https://doi.org/10.1073/pnas.1617454114

- Mayorga, E., Aufdenkampe, A. K., Masiello, C. A., Krusche, A. V., Hedges, J. I., Quay, P. D., et al. (2005). Young organic matter as a source of carbon dioxide outgassing from Amazonian rivers. *Nature*, 436(7050), 538–541. https://doi.org/10.1038/nature03880
- Messager, M. L., Lehner, B., Grill, G., Nedeva, I., & Schmitt, O. (2016). Estimating the volume and age of water stored in global lakes using a geo-statistical approach. *Nature Communications*, 7(1), 13603. https://doi.org/10.1038/ncomms13603
- Mineau, M. M., Wollheim, W. M., Buffam, I., Findlay, S. E. G., Hall, R. O., Hotchkiss, E. R., et al. (2016). Dissolved organic carbon uptake in streams: A review and assessment of reach-scale measurements. *Journal of Geophysical Research: Biogeosciences*, 121(8), 2019–2029. https://doi.org/10.1002/2015jg003204
- Montagnes, D. J. S., Morgan, G., Bissinger, J. E., Atkinson, D., & Weisse, T. (2008). Short-term temperature change may impact freshwater carbon flux: A microbial perspective. *Global Change Biology*, 14(12), 2823–2838. https://doi.org/10.1111/j.1365-2486.2008.01700.x
- Munn, N. L., & Meyer, J. L. (1990). Habitat-specific solute retention in two small streams: An intersite comparison. *Ecology*, 71(6), 2069–2082. https://doi.org/10.2307/1938621
- Newcomer, M. E., Hubbard, S. S., Fleckenstein, J. H., Maier, U., Schmidt, C., Thullner, M., et al. (2018). Influence of hydrological perturbations and riverbed sediment characteristics on hyporheic zone respiration of CO₂ and N₂. *Journal of Geophysical Research: Biogeosciences*, 123(3), 902–922. https://doi.org/10.1002/2017jg004090
- Obertegger, U., Flaim, G., Braioni, M. G., Sommaruga, R., Corradini, F., & Borsato, A. (2007). Water residence time as a driving force of zooplankton structure and succession. Aquatic Sciences, 69(4), 575–583. https://doi.org/10.1007/s00027-007-0924-z
- Peierls, B. L., & Paerl, H. W. (2010). Temperature, organic matter, and the control of bacterioplankton in the Neuse River and Pamlico Sound estuarine system. Aquatic Microbial Ecology, 60(2), 139–149. https://doi.org/10.3354/ame01415
- Raymond, P. A., & Bauer, J. E. (2000). Bacterial consumption of DOC during transport through a temperate estuary. Aquatic Microbial Ecology, 22(1), 1–12. https://doi.org/10.3354/ame022001
- Raymond, P. A., Hartmann, J., Lauerwald, R., Sobek, S., McDonald, C., Hoover, M., et al. (2013). Global carbon dioxide emissions from inland waters. *Nature*, 503(7476), 355–359. https://doi.org/10.1038/nature12760
- Raymond, P. A., & Saiers, J. E. (2010). Event controlled DOC export from forested watersheds. *Biogeochemistry*, 100(1–3), 197–209. https:// doi.org/10.1007/s10533-010-9416-7
- Raymond, P. A., Saiers, J. E., & Sobczak, W. V. (2016). Hydrological and biogeochemical controls on watershed dissolved organic matter transport: Pulse-shunt concept. *Ecology*, 97(1), 5–16. https://doi.org/10.1890/14-1684.1
- Rehage, H., & Kind, M. (2021). The first Damköhler number and its importance for characterizing the influence of mixing on competitive chemical reactions. *Chemical Engineering Science*, 229, 116007. https://doi.org/10.1016/j.ces.2020.116007
- Richey, J. E., Melack, J. M., Aufdenkampe, A. K., Ballester, V. M., & Hess, L. L. (2002). Outgassing from Amazonian rivers and wetlands as a large tropical source of atmospheric CO₂. Nature, 416(6881), 617–620. https://doi.org/10.1038/416617a
- Stream Solute Workshop. (1990). Concepts and methods for assessing solute dynamics in stream ecosystems. Journal of the North American Benthological Society, 9(2), 95–119. https://doi.org/10.2307/1467445
- United Nations Environment Programme. (2019). GEMStat Database of the Global Environment Monitoring System for Freshwater (GEMS/ Water) Programme (International Centre for Water Resources and Global Change (Ed.), Retrieved from https://gemstat.org/
- Vollenweider, R. A. (1975). Input-output models. Schweizerische Zeitschrift für Hydrologie, 37(1), 53-84. https://doi.org/10.1007/bf02505178
- Vörösmarty, C. J., Sharma, K. P., Fekete, B. M., Copeland, A. H., Holden, J., Marble, J., & Lough, J. A. (1997). The storage and aging of continental runoff in large reservoir systems of the world. *Ambio*, 26(4), 210–219.
- Ward, N. D., Keil, R. G., Medeiros, P. M., Brito, D. C., Cunha, A. C., Dittmar, T., et al. (2013). Degradation of terrestrially derived macromolecules in the Amazon River. *Nature Geoscience*, 6(7), 530–533. https://doi.org/10.1038/ngeo1817
- Wickland, K. P., Aiken, G. R., Butler, K., Dornblaser, M. M., Spencer, R. G. M., & Striegl, R. G. (2012). Biodegradability of dissolved organic carbon in the Yukon River and its tributaries: Seasonality and importance of inorganic nitrogen. *Global Biogeochemical Cycles*, 26(4), 2012GB004342. https://doi.org/10.1029/2012gb004342
- Wilson, H., Saiers, J., Raymond, P., & Sobczak, W. (2013). Hydrologic drivers and seasonality of dissolved organic carbon concentration, nitrogen content, bioavailability, and export in a forested New England stream. *Ecosystems*, 16(4), 604–616. https://doi.org/10.1007/s10021-013-9635-6
- Wollheim, W. M., Bernal, S., Burns, D. A., Czuba, J. A., Driscoll, C. T., Hansen, A. T., et al. (2018). River network saturation concept: Factors influencing the balance of biogeochemical supply and demand of river networks. *Biogeochemistry*, 141(3), 503–521. https://doi.org/10.1007/ s10533-018-0488-0
- Wollheim, W. M., Stewart, R. J., Aiken, G. R., Butler, K. D., Morse, N. B., & Salisbury, J. (2015). Removal of terrestrial DOC in aquatic ecosystems of a temperate river network. *Geophysical Research Letters*, 42(16), 6671–6679. https://doi.org/10.1002/2015gl064647
- Wollheim, W. M., Vörösmarty, C. J., Peterson, B. J., Seitzinger, S. P., & Hopkinson, C. S. (2006). Relationship between river size and nutrient removal. *Geophysical Research Letters*, 33(6), L06410. https://doi.org/10.1029/2006gl025845
- Worrall, F., Howden, N. J. K., & Burt, T. P. (2014). A method of estimating in-stream residence time of water in rivers. *Journal of Hydrology*, 512, 274–284. https://doi.org/10.1016/j.jhydrol.2014.02.050
- Yoon, B., & Raymond, P. A. (2012). Dissolved organic matter export from a forested watershed during Hurricane Irene. Geophysical Research Letters, 39(18). https://doi.org/10.1029/2012gl052785
- Yvon-Durocher, G., Allen, A. P., Montoya, J. M., Trimmer, M., & Woodward, G. (2010). The temperature dependence of the carbon cycle in aquatic ecosystems. In G. Woodward (Ed.), Advances in ecological research (pp. 267–313). Academic Press.
- Yvon-Durocher, G., Montoya, J. M., Trimmer, M., & Woodward, G. U. Y. (2011). Warming alters the size spectrum and shifts the distribution of biomass in freshwater ecosystems. *Global Change Biology*, 17(4), 1681–1694. https://doi.org/10.1111/j.1365-2486.2010.02321.x
- Zarfl, C., Lumsdon, A. E., Berlekamp, J., Tydecks, L., & Tockner, K. (2015). A global boom in hydropower dam construction. Aquatic Sciences, 77(1), 161–170. https://doi.org/10.1007/s00027-014-0377-0
- Zarnetske, J. P., Haggerty, R., Wondzell, S. M., & Baker, M. A. (2011). Dynamics of nitrate production and removal as a function of residence time in the hyporheic zone. *Journal of Geophysical Research*, 116(G1), 12. https://doi.org/10.1029/2010jg001356
- Zarnetske, J. P., Haggerty, R., Wondzell, S. M., Bokil, V. A., & González-Pinzón, R. (2012). Coupled transport and reaction kinetics control the nitrate source-sink function of hyporheic zones. Water Resources Research, 48(11). https://doi.org/10.1029/2012wr011894

References From the Supporting Information

Amon, R. M., & Benner, R. (1996). Bacterial utilization of different size classes of dissolved organic matter. *Limnology & Oceanography*, 41(1), 41–51. https://doi.org/10.4319/lo.1996.41.1.0041

- Bechtold, H. A., Marcarelli, A. M., Baxter, C. V., & Inouye, R. S. (2012). Effects of N, P, and organic carbon on stream biofilm nutrient limitation and uptake in a semi-arid watershed. *Limnology & Oceanography*, 57(5), 1544–1554. https://doi.org/10.4319/lo.2012.57.5.1544
- Berggren, M., Laudon, H., Jonsson, A., & Jansson, M. (2010). Nutrient constraints on metabolism affect the temperature regulation of aquatic bacterial growth efficiency. *Microbial Ecology*, 60(4), 894–902. https://doi.org/10.1007/s00248-010-9751-1

Bernhardt, E. S., & McDowell, W. H. (2008). Twenty years apart: Comparisons of DOM uptake during leaf leachate releases to Hubbard Brook Valley streams in 1979 versus 2000. *Journal of Geophysical Research*, 113(G3), G03032. https://doi.org/10.1029/2007jg000618

- Blaen, P. J., Milner, A. M., Hannah, D. M., Brittain, J. E., & Brown, L. E. (2014). Impact of changing hydrology on nutrient uptake in high arctic rivers. *River Research and Applications*, 30(9), 1073–1083. https://doi.org/10.1002/tra.2706
- Brookshire, E. N. J., Valett, H. M., Thomas, S. A., & Webster, J. R. (2005). Coupled cycling of dissolved organic nitrogen and carbon in a forest stream. *Ecology*, 86(9), 2487–2496. https://doi.org/10.1890/04-1184
- Catalán, N., Obrador, B., Felip, M., & Pretus, J. L. (2013). Higher reactivity of allochthonous vs. autochthonous DOC sources in a shallow lake. *Aquatic Sciences*, 75(4), 581–593. https://doi.org/10.1007/s00027-013-0302-y
- Cory, R. M., & Kaplan, L. A. (2012). Biological lability of streamwater fluorescent dissolved organic matter. *Limnology & Oceanography*, 57(5), 1347–1360. https://doi.org/10.4319/lo.2012.57.5.1347
- del Giorgio, P. A., & Pace, M. L. (2008). Relative independence of dissolved organic carbon transport and processing in a large temperate river: The Hudson River as both pipe and reactor. *Limnology & Oceanography*, 53, 185–197. https://doi.org/10.4319/10.2008.53.1.0185
- Dowell, W. H. M. (1985). Kinetics and mechanisms of dissolved organic carbon retention in a headwater stream. *Biogeochemistry*, 1(4), 329–352. https://doi.org/10.1007/bf02187376
- Guillemette, F., & del Giorgio, P. A. (2011). Reconstructing the various facets of dissolved organic carbon bioavailability in freshwater ecosystems. Limnology & Oceanography, 56(2), 734–748. https://doi.org/10.4319/lo.2011.56.2.0734
- Hall, R. O., Jr., & Meyer, J. L. (1998). The trophic significance of bacteria in a detritus-based stream food web. *Ecology*, 79(6), 1995–2012. https://doi.org/10.1890/0012-9658(1998)079[1995:ttsobi]2.0.co;2
- Hernes, P. J., & Benner, R. (2003). Photochemical and microbial degradation of dissolved lignin phenols: Implications for the fate of terrigenous dissolved organic matter in marine environments. Journal of Geophysical Research, 108(C9), 3291. https://doi.org/10.1029/2002jc001421
- Johnson, L. T., Tank, J. L., & Arango, C. P. (2009). The effect of land use on dissolved organic carbon and nitrogen uptake in streams. Freshwater Biology, 54(11), 2335–2350. https://doi.org/10.1111/j.1365-2427.2009.02261.x
- Kaplan, L. A., Wiegner, T. N., Newbold, J. D., Ostrom, P. H., & Gandhi, H. (2008). Untangling the complex issue of dissolved organic carbon uptake: A stable isotope approach. *Freshwater Biology*, 53(5), 855–864. https://doi.org/10.1111/j.1365-2427.2007.01941.x
- Köhler, S., Buffam, I., Jonsson, A., & Bishop, K. (2002). Photochemical and microbial processing of stream and soil water dissolved organic matter in a boreal forested catchment in northern Sweden. Aquatic Sciences, 64(3), 269–281. https://doi.org/10.1007/s00027-002-8071-z
- Langenheder, S., Kisand, V., Wikner, J., & Tranvik, L. J. (2003). Salinity as a structuring factor for the composition and performance of bacterioplankton degrading riverine DOC. FEMS Microbiology Ecology, 45(2), 189–202. https://doi.org/10.1016/s0168-6496(03)00149-1
- Lønborg, C., & Søndergaard, M. (2009). Microbial availability and degradation of dissolved organic carbon and nitrogen in two coastal areas. *Estuarine, Coastal and Shelf Science*, 81(4), 513–520. https://doi.org/10.1016/j.ecss.2008.12.009
- Miller, W. L., & Moran, M. A. (1997). Interaction of photochemical and microbial processes in the degradation of refractory dissolved organic matter from a coastal marine environment. *Limnology & Oceanography*, 42(6), 1317–1324. https://doi.org/10.4319/lo.1997.42.6.1317
- Mineau, M. M., Rigsby, C. M., Ely, D. T., Fernandez, I. J., Norton, S. A., Ohno, T., et al. (2013). Chronic catchment nitrogen enrichment and stoichiometric constraints on the bioavailability of dissolved organic matter from leaf leachate. *Freshwater Biology*, 58(2), 248–260. https:// doi.org/10.1111/fwb.12054
- Moran, M. A., Sheldon, W. M., Jr., & Zepp, R. G. (2000). Carbon loss and optical property changes during long-term photochemical and biological degradation of estuarine dissolved organic matter. *Limnology & Oceanography*, 45(6), 1254–1264. https://doi.org/10.4319/lo.2000.45.6.1254
- Newbold, J. D., Bott, T. L., Kaplan, L. A., Dow, C. L., Jackson, J. K., Aufdenkampe, A. K., et al. (2006). Uptake of nutrients and organic C in streams in New York City drinking-water-supply watersheds. *Journal of the North American Benthological Society*, 25(4), 998–1017. https:// doi.org/10.1899/0887-3593(2006)025[0998:uonaoc]2.0.co;2
- Oviedo-Vargas, D., Royer, T. V., & Johnson, L. T. (2013). Dissolved organic carbon manipulation reveals coupled cycling of carbon, nitrogen, and phosphorus in a nitrogen-rich stream. *Limnology & Oceanography*, 58(4), 1196–1206. https://doi.org/10.4319/lo.2013.58.4.1196
- Simon, K. S., Benfield, E. F., & Macko, S. A. (2003). Food web structure and the role of epilithic biofilms in cave streams. *Ecology*, 84(9), 2395–2406. https://doi.org/10.1890/02-334
- United Nations Environment Programme. (2019). GEMStat Database of the Global Environment Monitoring System for Freshwater (GEMS/ Water) Programme (International Centre for Water Resources and Global Change (Ed.), Retrieved from https://gemstat.org/